

**f** 

# **Earth's Future**

# **RESEARCH ARTICLE**

10.1029/2022EF003092

#### **Key Points:**

- Projections show lower peak snow water equivalent (SWE) below
  3,000 m and earlier peak SWE, peak surface water input (SWI) at all elevations
- Greater future peak SWI and reduced annual snow-derived SWI for all elevations, with a more even SWI distribution throughout the year
- A greater fraction of future SWE will be in high elevations that are currently unmonitored

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

J. C. Hammond, jhammond@usgs.gov

#### Citation:

Hammond, J. C., Sexstone, G. A., Putman, A. L., Barnhart, T. B., Rey, D. M., Driscoll, J. M., et al. (2023). High resolution SnowModel simulations reveal future elevation-dependent snow loss and earlier, flashier surface water input for the Upper Colorado River Basin. *Earth's Future*, *11*, e2022EF003092. https://doi. org/10.1029/2022EF003092

Received 8 AUG 2022 Accepted 26 DEC 2022

#### **Author Contributions:**

Conceptualization: John C. Hammond, Graham A. Sexstone Data curation: Graham A. Sexstone, Theodore B. Barnhart Formal analysis: John C. Hammond, Graham A. Sexstone, Annie L. Putman Funding acquisition: Graham A. Sexstone, David M. Rey, Jessica M. Driscoll

© 2023 The Authors. This article has been contributed to by U.S. Government employees and their work is in the public domain in the USA. This is an open access article under

the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# High Resolution SnowModel Simulations Reveal Future Elevation-Dependent Snow Loss and Earlier, Flashier Surface Water Input for the Upper Colorado River Basin

John C. Hammond<sup>1</sup>, Graham A. Sexstone<sup>2</sup>, Annie L. Putman<sup>3</sup>, Theodore B. Barnhart<sup>4</sup>, David M. Rey<sup>5</sup>, Jessica M. Driscoll<sup>6</sup>, Glen E. Liston<sup>7</sup>, Kristen L. Rasmussen<sup>8</sup>, Daniel McGrath<sup>9</sup>, Steven R. Fassnacht<sup>7,10</sup>, and Stephanie K. Kampf<sup>10</sup>

<sup>1</sup>Maryland-Delaware-D.C. Water Science Center, U.S. Geological Survey, Baltimore, MD, USA, <sup>2</sup>Colorado Water Science Center, U.S. Geological Survey, Lakewood, CO, USA, <sup>3</sup>Utah Water Science Center, U.S. Geological Survey, Salt Lake City, UT, USA, <sup>4</sup>Wyoming-Montana Water Science Center, U.S. Geological Survey, Helena, MT, USA, <sup>5</sup>Water Mission Area Observing Systems Division, U.S. Geological Survey, Lakewood, CO, USA, <sup>6</sup>Rocky Mountain Region, U.S. Geological Survey, Lakewood, CO, USA, <sup>6</sup>Rocky Mountain Region, U.S. Geological Survey, Lakewood, CO, USA, <sup>6</sup>Rocky Mountain Region, U.S. Geological Survey, Lakewood, CO, USA, <sup>6</sup>Rocky Mountain Region, U.S. Geological Survey, Lakewood, CO, USA, <sup>6</sup>Rocky Mountain Region, U.S. Geological Survey, Collins, CO, USA, <sup>8</sup>Department of Atmospheric Science, Colorado State University, Fort Collins, CO, USA, <sup>9</sup>Department of Geosciences, Colorado State University, Fort Collins, CO, USA, <sup>10</sup>Department of Ecosystem Science and Sustainability, Colorado State University, Fort Collins, CO, USA

Abstract Continued climate warming is reducing seasonal snowpacks in the western United States, where >50% of historical water supplies were snowmelt-derived. In the Upper Colorado River Basin, declining snow water equivalent (SWE) and altered surface water input (SWI, rainfall and snowmelt available to enter the soil) timing and magnitude affect streamflow generation and water availability. To adapt effectively to future conditions, we need to understand current spatiotemporal distributions of SWE and SWI and how they may change in future decades. We developed 100-m SnowModel simulations for water years 2001-2013 and two scenarios: control (CTL) and pseudo-global-warming (PGW). The PGW fraction of precipitation falling as snow was lower relative to CTL, except for November-April at high elevations. PGW peak SWE was lower for low (-45%) and mid elevations (-14%), while the date of peak SWE was uniformly earlier in the year for all elevations (17-23 days). Currently unmonitored high elevation snow represented a greater fraction of total PGW SWE. PGW peak daily SWI was higher for all elevations (30%-42%), while the dates of SWI peaks and centroids were earlier in the year for all elevations under PGW. PGW displayed elevated winter SWI, lower summer SWI, and changes in spring SWI timing were elevation-dependent. Although PGW peak SWI was elevated and earlier compared to CTL, SWI was more evenly distributed throughout the year for PGW. These simulated shifts in the timing and magnitude of SWE and SWI have broad implications for water management in dry, snow-dominated regions.

**Plain Language Summary** Snowpack water storage has historically functioned as a reliable extension of manmade reservoir storage. Loss of this storage has consequences for water resource management, ecological communities, and natural hazards including wildfire. We modeled snow accumulation and melt at high spatial resolution in the Upper Colorado River Basin to assess patterns in the timing and magnitude of snow storage and snowmelt for historical and future scenarios. We analyze these patterns in relation to existing snow monitoring station coverage, and ask how this coverage may need to change in future decades to better represent water availability. Our results indicate widespread future snow storage losses at lower elevations, but limited change at higher elevations that will likely remain conducive to seasonal snow accumulation and melt for decades to come. Peak snow storage and peak snowmelt occurred earlier for all elevations in future years, with increased peak surface water input noted at all elevations. A greater fraction of future snow storage will be in currently unmonitored high elevations. Projected elevation dependent changes from this study have implications for other dry, snow dominated regions, and additional work is needed to evaluate combined effects of widespread snow loss and earlier, flashier input on coordinated water management.

# 1. Introduction

Globally, snow loss is causing cascading impacts to soil storage, evapotranspiration, streamflow and groundwater recharge; sediment transport and hazards; aquatic and terrestrial ecology; as well as human water use for public



Graham A. Sexstone

Driscoll

Liston

Driscoll

L. Putman

M. Rev

Investigation: John C. Hammond,

Methodology: John C. Hammond,

Graham A. Sexstone, Annie L. Putman, Jessica M. Driscoll, Glen E. Liston

Project Administration: Graham A.

Sexstone, David M. Rey, Jessica M.

Software: Graham A. Sexstone,

Theodore B. Barnhart, Jessica M. Driscoll, Glen E. Liston

Validation: John C. Hammond

Writing - original draft: John C.

Resources: Jessica M. Driscoll, Glen E.

Supervision: John C. Hammond, Graham A. Sexstone, David M. Rey, Jessica M.

Visualization: John C. Hammond, Annie

Hammond, Graham A. Sexstone, Annie

L. Putman, Theodore B. Barnhart, David

Writing - review & editing: John C.

Hammond, Graham A. Sexstone, Annie

L. Putman, Theodore B. Barnhart, David M. Rey, Glen E. Liston, Kristen L.

Rasmussen, Daniel McGrath, Steven R.

Fassnacht, Stephanie K. Kampf

supply, irrigation, mining, hydropower, agriculture, and recreation (e.g., Huss et al., 2017). Continued global warming is reducing seasonal snowpacks in the western United States (Mote et al., 2018), where a majority of the region's water supply has historically been generated by snowmelt, especially in semi-arid basins (Li et al., 2017). These changes are modifying the timing and magnitude of streamflow, with (a) mid-winter melt occurring more frequently, (b) declines in peak snow water equivalent (SWE) (Musselman et al., 2021), and (c) earlier streamflow timing in snowmelt driven areas (Dudley et al., 2017; Stewart et al., 2005). Across the western United States, SWE declines of  $\sim 25\%$  are expected by the middle of the 21st century, and seasonal snowpacks are likely to be replaced by low-to-no snow conditions at low and mid elevations (Fyfe et al., 2017; Siirila-Woodburn et al., 2021).

Snowmelt in the Upper Colorado River Basin (UCRB) generates as much as 92% of the streamflow for the larger Colorado River Basin, which supports over 40 million people and irrigates 5.5 million acres across seven western U.S. states (Butler et al., 2015; Lukas & Payton, 2020). Thus, better understanding the present spatiotemporal distribution of SWE and the associated hydrologic response to global warming in this region is critical to prepare for future adjustments to water resource management. Existing allocations between upper and lower basin states may be unrealizable with continued snow loss and heightened interannual climatic variability, especially for persistent drought periods (Adler, 2008; Udall & Overpeck, 2017; Woodhouse et al., 2006). Prior work in the Colorado River Basin has shown annual streamflow to be most sensitive to warming during the warm season as compared to other seasons for the last several decades (Ban & Lettenmaier, 2022), with annual discharge decreasing 9.3% per °C of warming over the period 1913–2017 (Milly & Dunne, 2020). Other recent studies report that increases of temperature of 1-2°C in the Colorado River Basin without changes in precipitation could result in streamflow reductions of 10%-20% (Christensen et al., 2004; Christensen & Lettenmaier, 2007; McCabe and Wolock, 2007, 2011). Projections using regional climate models indicate reduced, earlier peak SWE in the UCRB (Rasmussen, Baker et al., 2011) and increases in streamflow for the 2030s followed by decreases for the 2080s compared to present, with potential streamflow increases insufficient to meet growing water demand (Miller et al., 2021).

Accurately characterizing mountain precipitation and SWE remains a key task for responding to projected streamflow reductions and intensified droughts (Kim et al., 2021). This characterization has been a challenge because most global climate models (GCMs) or mesoscale climate models produce simulations at coarse spatial resolutions and are unable to represent complex orographic precipitation patterns and small-scale snow processes that are important in mountainous regions such as the UCRB, resulting in substantial biases (Salzmann & Mearns, 2012). Fine spatial scale is crucial for regional predictions of future hydroclimate (Ikeda et al., 2021; Wrzesien & Pavelsky, 2020) to accurately represent the influence of topographic complexity on orographic precipitation patterns and snow process dynamics (Liston, 2004; López-Moreno et al., 2013; Musselman, Clark, et al., 2017, Musselmann, Molotch, & Margulis et al., 2017; Rasmussen, Baker et al., 2011). In this study, we extend prior mesoscale snow modeling work for areas of the western United States (e.g., Ikeda et al., 2021; Li et al., 2017; Musselman, Clark, et al., 2017) to a finer resolution by using process-based snow modeling (Snow-Model; Liston & Elder, 2006a) at 100-m (m) spatial resolution driven with fine-resolution convective-permitting and orography-resolving Weather Research and Forecasting Model (WRF) simulations for a control (CTL) and a pseudo-global-warming (PGW) scenario. The PGW method (Liu et al., 2017; Rasmussen, Baker et al., 2011, Rasmussen, Liu et al., 2011; Sato et al., 2007; Schär et al., 1996) is best used to ask the question, "what will today's weather look like in a future climate state." The difference between two 30-year monthly mean conditions in a future (2071-2100) and current (1976-2005) are applied to reanalysis data used to drive the mesoscale model that allows for historical events and patterns (i.e., extreme storm events, dry vs. wet seasons) to be directly studied in contrast to other methods that dynamically downscale from GCMs. The approach well characterizes the thermodynamic aspects of climate change, but does not represent large-scale changes in atmospheric circulation. However, sub-monthly mesoscale variability is well captured and allows for significant improvements in the representation of precipitating systems in regions of complex terrain (Rasmussen, Baker et al., 2011, Rasmussen & Liu, 2017).

This study optimally combines two cutting-edge models at a fine spatial scale that better captures the effects of complex topography and elevation gradients on snow variability and snow processes than prior studies at 4-km or coarser resolutions, and it allows us to investigate how snowpack processes change with climate warming without introducing uncertainty inherent to other methods that include potential changes in storm tracks. Additionally, in this study we analyze surface water input (SWI), which is the sum of rainfall and snowmelt

available to enter the soil, to characterize the effects of a changing snowpack on the hydrologic cycle. Prior studies at plot and catchment scale have demonstrated linkages between SWI and observed and modeled streamflow (Hammond et al., 2019; Kiewiet et al., 2022; Kormos et al., 2014), and the use of SWI in this study enables linking snow changes to hydrologic effects not fully captured in prior studies using metrics including snowfall fraction (Berghuijs et al., 2014) and snowmelt rate (Barnhart et al., 2016).

With the aim to reduce uncertainty in present and future spatiotemporal distributions of SWE and SWI, we ask our primary study question (a) how will SWE change under a future warmer climate by elevation, and how might this affect future water availability via changes to the timing and magnitude of SWI? (Sections 3.3, 3.4, 4.1). Using PGW as an indicator of potential future conditions associated with the thermodynamic aspects of climate change, we hypothesize future SWI will occur earlier and at lower magnitude for mid to low elevations coincident with decreases in SWE (Barnhart et al., 2016; Musselmann, Molotch, & Margulis et al., 2017), but peak SWE may increase at times for high elevations where precipitation is expected to increase; temperatures during late fall, winter and early spring will still be conducive for snowfall (e.g., Musselman, Clark, et al., 2017). Despite similar or elevated SWE at high elevations, we hypothesize the timing of future SWI will be earlier in the year than for historical estimates (Musselmann, Molotch, & Margulis et al., 2017). To place these results in context of current snow monitoring locations following in depth analysis of the present and future distributions of SWE and SWI, we ask a secondary study question (b) how well do current snow observation stations represent the spatial distribution of SWE and SWI, and how might these networks need to change to continue to provide predictive information for future water availability? (Section 4.2). We hypothesize future snowpacks will be more limited to higher elevations than at present (Ikeda et al., 2021), and that low to mid elevations will provide a smaller snowmelt contribution to streamflow despite substantial present-day snowmelt contribution given their large spatial footprint (Hammond et al., 2018; Harrison et al., 2021).

# 2. Methods

To answer our study questions, we selected the Colorado portion of the UCRB (Section 2.1), implemented CTL and PGW SnowModel simulations (Section 2.2), and performed a multi-source model evaluation using station and remotely sensed data (Section 2.3). We then aggregated model inputs and outputs to elevation bands and daily, monthly, and annual timesteps, and calculated several annual metrics to describe differences in SWE and SWI between CTL and PGW scenarios (Section 2.4).

## 2.1. Study Area

The model domain was a  $311 \times 300$  km area in Colorado, United States, encompassing the headwaters of Colorado and Gunnison River Basin portions of the UCRB (Figure 1). Elevations within the domain ranged from 1,313 to 4,385 m (Figure 1). The model domain is a snow-dominated mountainous region with deep persistent snowpacks and high runoff generation (Figure 1; Hammond et al., 2018; Miller et al., 2016); seasonal snowpacks in the region serve as the primary source of annual streamflow with snowmelt contributing to an estimated 71% of total UCRB runoff (Li et al., 2017).

#### 2.2. Model Description and Simulations

To simulate detailed snow evolution processes for both current and future climate conditions, we utilized Snow-Model, a spatially distributed physics-based snow evolution modeling system (Liston & Elder, 2006a; Liston et al., 2020), forced with WRF convection-permitting and orography-resolving regional climate simulations on a 4-km grid resolution (Liu et al., 2017; Rasmussen & Liu, 2017). WRF climate simulations provided the atmospheric forcing conditions to drive SnowModel in both a current and future climate scenario. A pair of continuous 13-water-year (2001–2013) WRF model simulations was used: (a) a current climate simulation CTL forced using ERA-Interim reanalysis, and (b) a future climate simulation using the PGW method. The PGW simulation uses the ERA-Interim reanalysis for the same period as (a) and adds an ensemble mean climate delta (increase in radiative forcing) from 100 years in the future for the most extreme Representative Concentration Pathway (RCP) 8.5 scenario. The two-dimensional 4-km grid spacing hourly WRF simulation output (Rasmussen & Liu, 2017) for both CTL and PGW scenarios were subset to our study area model domain and used as atmospheric forcing for running SnowModel.





**Figure 1.** Snow modeling study area with (a) elevation at 100-m modeling resolution and existing snow telemetry stations and (b) Moderate Resolution Imaging Spectroradiometer snow persistence (SP, the % of time snow is present from 1 January to 3 July) at 500-m resolution. Inset in (a) shows elevation distribution, inset in (b) shows cumulative fraction of the study area. (c) Location of the snow modeling domain study area and the Upper Colorado River Basin in the United States. Black and orange contours in (a, b) and the vertical lines of the same color in the insets of (a and b) separate the domain into three elevational categories that are used for synthesizing patterns throughout our analyses: "high," >3,300 m; "mid," 2,300 m < elevation  $\leq$  3,300 m; "low," 2,300 m.

SnowModel simulations were run for both CTL and PGW scenarios with a three-hour time step and 100-m spatial grid resolution (9.32 million grid cells across the model domain; Figure 1; Sexstone et al., 2022). Elevation and land cover data (30-m spatial resolution) were provided by the U.S. Geological Survey (USGS) National Elevation Data set (U.S. Geological Survey, 2020) and North American Land Change Monitoring System (Canada Centre for Remote Sensing, 2020) and were resampled to the 100-m spatial grid resolution using average resampling and mode resampling (GDAL/OGR, 2022), respectively. Hourly WRF precipitation, air temperature, relative humidity, and wind speed and direction data were aggregated to three-hour values to correspond with the model simulation time step. Three-hour WRF forcing data along with the mean elevation of each 4-km WRF grid cell were used by MicroMet (Liston & Elder, 2006b), a high-resolution meteorological distribution submodel of SnowModel, to downscale and create the 100-m spatial resolution meteorological forcing data required to run SnowModel. The Dai (2008) rain-snow precipitation phase fraction parameterization (Liston et al., 2020) was used to calculate a rain-snow fraction of precipitation inputs for this study. Other submodels that make up the SnowModel modeling system were used to simulate spatially distributed snowpack evolution across the study area including: EnBal (Liston, 1995), which computes surface energy exchanges between the snow and atmosphere; SnowPack (Liston & Hall, 1995), which simulates the seasonal evolution of snow depth and SWE; and SnowTran-3D (Liston & Sturm, 1998), a three-dimension model that simulates snow redistribution by wind over topographically variable terrain. CTL and PGW SnowModel simulations include snow sublimation from the surface, canopy, and blowing snow (Sexstone et al., 2018). SnowModel simulation outputs (100-m spatial resolution) of SWE, SWI, snowmelt, precipitation, fraction of precipitation falling as snow, and air temperature were aggregated from a three-hour to a daily time step and used in this study.

#### 2.3. Model Evaluation

SnowModel simulations for the CTL scenario were evaluated using two independent data streams, snow telemetry (SNOTEL) station observations and Moderate Resolution Imaging Spectroradiometer (MODIS) satellite observations. Simulated precipitation and SWE values were evaluated using SNOTEL station observations of precipitation and SWE at 71 sites ranging in elevation from 2,591 to 3,523 m that included at least three complete water years of observations during the study period (Table S1 in Supporting Information S1). Simulated snow-covered

area and snow persistence (SP) were evaluated using daily cloud gap filled MODIS observations, and derived SP grids. A grid cell and day in the model output was considered to be snow covered if SWE  $\geq 10$  mm; and SP was calculated as the fraction of time snow was present from 1 January to 3 July. Evaluation using MODIS data was completed for all areas of the domain in the seasonal snow zone, where SP is greater than 50% (Figure S3 in Supporting Information S1). Prior work has evaluated the WRF forcing (Rasmussen & Liu, 2017) used in our model simulations, and the model estimates reliably reproduced the annual, seasonal, and sub-seasonal precipitation and surface temperature climatology in mountain regions of the western United States (Liu et al., 2017; Rasmussen, Baker et al., 2011). Likewise, SnowModel has been rigorously evaluated and shown to perform well in seasonally snow-covered environments similar to the study area (e.g., Greene et al., 1999; Hiemstra et al., 2006; Liston & Elder, 2006a, 2006b; Liston & Hiemstra, 2008; Prasad et al., 2001; Sproles et al., 2013; Sexstone et al., 2018; Sexstone, Penn et al., 2020). For details on the methods used to evaluate the SnowModel simulations used in this study, please refer to Text S1 in Supporting Information S1.

#### 2.4. Analysis of Model Inputs and Outputs

We examined the precipitation, temperature, and fraction of precipitation falling as snow on monthly, annual, and mean annual timescales across 13 years of simulation. We further examined SWE and SWI at daily resolution. We focus many of our analyses on variability in mean annual SWE and SWI by elevation, as prior studies have used coarser spatial resolutions less able to resolve mountain topography and elevational snow variability (Ikeda et al., 2021; Musselmann, Molotch, & Margulis et al., 2017). Because the 13-year window is too short for evaluating temporal variability and trends, we evaluated 13-year average differences between CTL and PGW scenarios. We aggregated daily SWE and SWI to 100-m elevation bands (Figure S1 in Supporting Information S1) using the *exactextractr* (Baston, 2020) R package to obtain daily area-weighted averages for each unit. Spatial units across the domain were classified into 31,100-m elevation bands ranging from 1,300 to 4,300 m. For mean annual scale analyses, the mean of annual totals (e.g., precipitation) or the mean of annual means (e.g., temperature) were computed. For analyses plotting results by elevation, we computed the median value of a given metric for all spatial units falling within each 100-m elevation band. For example, when displaying the CTL peak SWE for 3,000 m in Figure 3, we first calculated the peak SWE for each simulation year (2001–2013) and each spatial unit, then took the mean of all simulation years for that unit, and finally took the median value of peak SWE for all spatial units with the 100-m elevation classification of 3,000 m.

We used a threshold in rainfall/SWE to differentiate between snowmelt dominated input and mixed input because without a threshold, snowmelt events with an insignificant amount of rain would be categorized as mixed. A 10% threshold in rainfall/SWE, where input with rainfall/SWE >= 10% is mixed input and rainfall/SWE < 10% is snowmelt input, was adopted here and is functionally similar to those used in other rain on snow studies (Cohen et al., 2015; Guan et al., 2016; Lundquist et al., 2008). We calculated six annual SWE and SWI metrics for each spatial unit (Figure S1 in Supporting Information S1) using median daily values of model outputs as described above. Our focus is on SWI, which includes rainfall and snowmelt, because it is more indicative of hydrologic response in snow dominated areas than precipitation alone. Annual metrics calculated were: (a) peak SWI, which represents peak daily SWI, (b) peak SWE, (c) peak SWI DOWY, the day of the water year on which the peak SWI occurs, (d) peak SWE DOWY, the day of the water year when peak SWE occurs, (e) SWI50 DOWY, the day of the water year on which 50% of the annual SWI has occurred, and (f) input seasonality (*IS*), which is similar to the precipitation seasonality metric (Dingman, 2002; Markham, 1970) but calculated with SWI time series rather than precipitation (Barnhart et al., 2022). *IS* is an index of how much mean annual input occurs within a concentrated period or is equally distributed throughout the year and is calculated using monthly SWI totals.

$$IS = \frac{1}{A} \left( \sum_{i=1}^{12} \left| M_i - \frac{A}{12} \right| \right)$$
(1)

where:

A is the annual sum of SWI.

 $M_i$  is the monthly sum of SWI.

Finally, we computed the fraction of SWI sourced from snowmelt  $(f_{snow})$  and mixed input  $(f_{mix})$  for each day using summed total values of SWI for each spatial unit (rather than mean values) separated into three categories: snowmelt

input (rainfall/SWE  $\leq 0.1$  and SWE > 0), mixed input (rainfall/SWE > 0.1 and SWE > 0), and rainfall input (rainfall > 0 and SWE = 0), and these values were then totaled for each spatial unit to calculate  $f_{snow}$  and  $f_{mix}$  as:

$$f_{\text{snow}} = \left(\frac{\text{snowmelt input}}{\text{snowmelt input} + \text{rainfall input} + \text{mixed input}}\right)$$
(2)  
$$f_{\text{mix}} = \left(\frac{\text{mixed input}}{\text{snowmelt input} + \text{rainfall input} + \text{mixed input}}\right)$$
(3)

In addition to assessing SWI changes by elevation between CTL and PGW scenarios using annual scale metrics, we implemented two subsequent analyses to examine domain-wide impacts to total SWI. To better understand patterns and changes in total study area mean annual SWI by elevation and time of the year, we calculated the fraction of total study area mean annual SWI generated by each elevation band for each month for CTL and PGW separately. To investigate snowmelt contributions to SWI, and how they differ between PGW and CTL, we mapped the mean annual fraction of SWI contributed from snowmelt and calculated the mean monthly snowmelt contributed SWI for both scenarios, and how this changed between scenarios. Finally, we investigated differences between mean annual change and annual change within the same scenario and between scenarios for two simulation years (2011, greatest peak SWE for CTL, and 2012, lowest peak SWE for CTL) that highlight high interannual variability of snow and SWI.

In reporting the results of our analyses, we use three elevation categories to synthesize patterns in SWE and SWI change by elevation: "high," >3,300 m; "mid," 2,300 m < elevation  $\leq$  3,300 m; "low,"  $\leq$ 2,300 m. These elevation categories were chosen because they broadly reflect the satellite-derived boundaries of the present intermittent and seasonal snow zones in the study region, zones that reflect similar snow patterns (Moore et al., 2015) and similar patterns of soil moisture and streamflow response to snowmelt (Hammond et al., 2018; Harrison et al., 2021). These elevation categories are also consistent with regional regression equation elevation ranges for streamflow prediction in ungaged areas (Capesius & Stephens, 2009; Eurich et al., 2021).

# 3. Results

To examine differences in SWE and SWI between CTL and PGW scenarios, we first evaluated performance of SnowModel in the Colorado portion of the UCRB for the CTL scenario (Section 3.1) and assessed differences in the meteorological forcing between CTL and PGW (Section 3.2). We then quantified differences in SWE (Section 3.3) and SWI (Section 3.4) between CTL and PGW using daily model output and annual-scale synthesis metrics for elevation bands across the study area.

## 3.1. Model Evaluation

We evaluated simulated SnowModel SWE for the historic (water years 2001-2013) CTL scenario using SNOTEL stations observations (Figure S2, Table S1 in Supporting Information S1) and simulated SnowModel snow covered area using MODIS observed snow covered area for water years 2001-2005. Comparison of simulated SWE with all available observations from 71 SNOTEL stations within the study domain showed a root mean square error (RMSE) of 98 mm and Nash-Sutcliffe efficiency (NSE) of 0.72 for periods when both observed and simulated SWE were not equal to zero. NSE performance statistics computed individually for simulation results at each SNOTEL station (Table S1 in Supporting Information S1) show 30% of stations displayed a very good performance rating (0.75 < NSE < 1.00), 25% of stations had good performance (0.65 < NSE < 0.75), 17% of stations were satisfactory (0.50 < NSE < 0.65), and 28% of stations were unsatisfactory (NSE < 0.50) according to criteria established in Moriasi et al. (2007). Simulated SWE at SNOTEL stations was biased low overall (PBias -6.7%, mean error -14 mm) compared to observations, following a low bias in simulated precipitation (i.e., WRF precipitation downscaled to a 100 m spatial resolution by SnowModel) relative to observations (PBias -6.5%, mean error -27 mm). Simulated precipitation exhibited an RMSE of 97 mm and NSE of 0.88 when compared to SNOTEL observations. The SnowModel snow covered area daily time series by elevation band showed agreement with remotely sensed values with a median study area NSE of 0.33 and PBias of 1.9% for areas with seasonal snowpack (SP > 50%). SnowModel differed most from remotely sensed estimates during abrupt accumulation and melt events, but showed close alignment for accumulation and melt periods (Figure S3a-d in Supporting Information S1). The NSE of modeled and remotely sensed snow-covered area values for 100-m





Figure 2. Mean monthly (a) precipitation, (b) temperature, (c) snowfall fraction of precipitation, and (d) surface water input difference between pseudo-global-warming (PGW) and control (CTL) scenarios for 100-m elevation bands across the study area. In all cases values presented are PGW minus CTL.

elevation band units across the study area, performance was generally best for mid and high elevation areas in the seasonal snow zone (Figure S3e in Supporting Information S1), with poorer performance in low elevation, intermittently snowy areas. Finally, in comparison to existing SWE products at 4-km and 1-km spatial resolutions, the 100-m simulations for current climate conditions used in this paper better matched the spatial variability of near peak SWE for the UCRB (Figures S17, S18 in Supporting Information S1). These results suggest the forcing data and level of performance of SnowModel is adequate for our analyses tracking snow accumulation and melt.

#### 3.2. Differences in Meteorological Forcings Between Scenarios

The PGW scenario was warmer and wetter than the CTL scenario and exhibited reductions to the fraction of precipitation falling as snow. Mean monthly PGW temperatures for all months ranged from 4.9 to 5.6°C warmer than for CTL for elevation bands across the domain. We observed the greatest positive changes in temperature in the eastern, high elevation region of the study domain, with the greatest changes in fall (Figure 2b), followed by spring. In these seasons, high elevations exhibited the greatest positive changes, with changes in October up to 6.3°C for high elevations. The smallest long-term changes occurred during winter on the order of 4°C.

Following the thermodynamic relation between temperature and specific humidity, mean monthly PGW precipitation tended to be higher, by approximately 146 mm/y relative to CTL across all elevations. The greatest precipitation changes occurred for high elevations in the central and northeast parts of the study domain. Precipitation was higher compared with CTL in all seasons except spring (Figure 2a). The greatest positive precipitation changes occurred during winter at high elevations, resulting in increased snowfall, and the greatest negative changes occurred during spring months at high elevations.

The fraction of precipitation delivered as snow was reduced by as much as 40% between CTL and PGW, with every 100-m elevation band indicating lower snowfall fraction during at least one month (Figure 2c). In cold season months between November and April, high elevations indicated little change in the fraction of precipitation delivered as snow, whereas the low elevations indicated losses of up to 40%, especially in December through February. High elevations indicated the greatest losses in proportion of precipitation delivered as snow during May and October, likely owing to positive changes in temperature projected during those months (e.g., Klos et al., 2014; Scalzitti et al., 2016a, 2016b).



**Figure 3.** Mean annual averages of surface water input (SWI) and snow water equivalent (SWE) annual metrics by elevation and scenario (control - CTL, pseudo-global-warming - PGW) as well as values for 2011 (high snow year) and 2012 (low snow year). Figure S9 in Supporting Information S1 contains annual SWI metrics calculated using only snowmelt-derived SWI. Transparent orange, yellow and blue coloring in the background of each panel reflects three elevation categories used to synthesize patterns in snow and SWI changes by elevation: high, >3,300 m; mid, 2,300 m < elevation  $\leq$  3,300 m; low,  $\leq$ 2,300 m.

## 3.3. Differences in SWE and Snowmelt Between Scenarios

#### **3.3.1. High Elevations**

We evaluated changes in the timing and magnitude of SWE between the CTL and PGW scenarios using annual metrics that synthesize relevant shifts in daily time series of SWE. The greatest mean annual peak SWE for both scenarios occurred at the highest elevations (Figure 3), with the PGW scenario exhibiting, on average, peak SWE values that were only 3% lower (13 mm) relative to CTL. Despite higher mean annual precipitation for PGW at all elevations during most of the snow accumulation season (Figure 2), peak SWE was only higher over a small, high elevation area in the PGW scenario relative to CTL (Figure 3, Figure S13 in Supporting Information S1). The date of peak SWE was latest at the high elevations for CTL and earlier at high elevations in the PGW scenario (-17 days, Table S4 in Supporting Information S1). Additional comparisons using mean daily values (Figure 4) and mean monthly values (Table S2 in Supporting Information S1) confirmed SWE magnitude and timing changes observed using annual metrics. The months where high-elevation SWE differed most between the scenarios were May-July (-44 to -191 mm, Table S2 in Supporting Information S1) and October-December (-21 to -34 mm). On average, during these months, the PGW scenario exhibited lower SWE than the CTL scenario. High elevations showed elevated February to April SWE from CTL to PGW (+1 to +13 mm, Table S2 in Supporting Information S1), with patterns in average monthly snowmelt matching the patterns observed for SWE change between CTL and PGW. Notably, while snow did not melt at high elevations during the CTL scenario for the months December to February, small amounts of snowmelt were generated for these months in the PGW scenario (Table S2 in Supporting Information S1).

#### 3.3.2. Mid Elevations

When PGW was compared with CTL, pronounced reductions in peak SWE occurred at mid elevations (-26 mm, -14%, Table S4 in Supporting Information S1, Figure 3). The date of peak SWE was earlier during the water year (-23 days, Table S4 in Supporting Information S1), a larger temporal shift than observed for high or low elevations. Changes in mean monthly values revealed that mid elevations displayed the largest reductions in SWE for April (-53 mm, Figure S15 in Supporting Information S1) and May (-51 mm, Table S2 in Supporting Information S1), and exhibited no change for the months July–September when snow was absent in both scenarios. Monthly mean snowmelt was higher for the mid elevations during winter months, as earlier peak SWE corresponded with greater winter melt and reduced spring and summer snowmelt (Table S2, Figure S8 in Supporting Information S1).

#### 3.3.3. Low Elevations

Low elevations displayed the greatest percentage reductions in peak SWE for the PGW scenario compared to CTL (-30 mm, -45%, Table S4 in Supporting Information S1) and when comparing low elevation reductions to those at mid and high elevations. The date of peak SWE occurred 21 days earlier, a similar advancement to the earlier peak SWE observed for mid elevations. Low elevations displayed reductions in mean monthly SWE (-60% to -100%, Table S2 in Supporting Information S1) for all months except for July and August which lacked SWE in both scenarios. These reductions were further highlighted by patterns displayed using mean daily SWE values for the two scenarios (Figure 4). Although all elevations experienced earlier peak SWE and subsequently snowmelt in the PGW scenario (Figure 4, Figure S8, Table S2 in Supporting Information S1).

#### 3.4. Differences in Surface Water Input Between Scenarios

#### **3.4.1. High Elevations**

Because we observed differences in SWE and snowmelt between the two scenarios, we then analyzed differences in SWI between CTL and PGW scenarios by elevation. Mean annual peak daily SWI was highest in the high elevation band for CTL (Figure 3), with higher peak daily SWI in the PGW scenario compared to CTL (+30%, +12 mm/d, Table S4 in Supporting Information S1). The date of peak daily SWI for high elevations occurred 14 days earlier in the PGW scenario. The IS metric, which indicates whether input is focused primarily within a few months or is evenly distributed throughout the year, showed that the most temporally concentrated input was at high elevations during CTL. That input became less concentrated for PGW at the same elevations (-13%, Table S4 in Supporting Information S1). For the CTL scenario, the mean annual centroid date of SWI was earlier in the western part of the domain, and later at high elevations and in the eastern part of the study area. The

# **Earth's Future**





**Figure 4.** Mean annual daily snow water equivalent (SWE, top) and mean annual daily surface water input (SWI, bottom) by scenario (control - CTL, pseudo-globalwarming - PGW) and elevation band. A 7-day moving mean was applied for better visualization between bands, but this compresses peak daily SWI. SWE and SWI patterns are separated into three elevational categories: high, >3,300 m; mid, 2,300 m < elevation  $\leq$  3,300 m; low,  $\leq$ 2,300 m. Figure S7 in Supporting Information S1 displays greater details in daily SWE and SWI for the low elevation category.

centroid date of mean annual SWI was earlier for PGW than for CTL for all elevations (Table S4, Figure S14 in Supporting Information S1).

Mean monthly SWI was higher in the PGW scenario than the CTL scenario for all months except for June and July (Table S3 in Supporting Information S1). The magnitude of the difference was greatest in April and May, with the negative differences in June and July nearly perfectly offsetting the positive differences in April and May from a mass balance perspective. Changes observed in mean monthly SWI were further evidenced by patterns in mean daily values. For CTL, high elevations experienced long and continuous periods of elevated snowmelt-derived





**Figure 5.** Time series of daily surface water input (SWI) by elevation colored by the fraction of SWI sourced from snowmelt ( $f_{snow}$ ) for control (CTL) and pseudo-global-warming (PGW) scenarios. A 7-day moving mean was applied for better visualization between bands, but this compresses peak daily SWI. SWI patterns are separated into three elevational categories: high, >3,300 m; mid, 2,300 m < elevation  $\le$  3,300 m; low,  $\le$ 2,300 m.

SWI in May and June (Figures 4 and 5), whereas PGW SWI at high elevations displayed higher, concentrated peaks earlier in the year. From here forward, we use the term "flashier" to represent more rapidly occurring and higher peak SWI observed for PGW as compared to CTL. At high elevations, snowmelt contributions to SWI for CTL were greatest in April and May (Figure 5), with PGW snowmelt contributions to SWI generally higher earlier in the spring and lower later in spring and summer.

#### 3.4.2. Mid Elevations

Peak daily SWI was similarly greater for PGW at mid elevations relative to CTL as compared to high elevations (+34%, +10 mm/d, Table S4 in Supporting Information S1). However, the date of peak daily SWI was 20 days earlier in the year at mid elevations, a greater advance than observed for high elevations. IS declined the most at mid and low elevations (-18%, Table S4 in Supporting Information S1). Mean monthly total SWI values for mid elevations revealed higher values in monthly SWI for October–April, similar to what was observed for high elevations, but with reductions observed for May, June, July, and September. Elevated SWI for March and April in mid elevations displayed the greatest reductions (Figure S15 in Supporting Information S1). Similar to the PGW-CTL differences observed in mean daily values for high elevations, mid elevations displayed elevated daily SWI (Figure 4) and earlier daily snowmelt contributions to SWI (Figure 5). This period was followed by reductions in SWI and snowmelt contributions, which occurred earlier in the year than at high elevations. While peak daily SWI for high elevations consistently displayed higher values for PGW, some mid elevation bands displayed lower and longer elevated periods potentially due to higher SWI earlier in the season (Figures 4 and 5).

#### 3.4.3. Low Elevations

At low elevations, peak daily SWI was higher for PGW, and with a greater percent difference than for mid and high elevations (+42%, +10 mm/d, Table S4 in Supporting Information S1). The date of peak daily SWI occurred 23 days earlier in the year and the centroid date of SWI occurred 30 days earlier in the year. Both metrics indicate greater differences between scenarios than at mid or high elevations. At low elevations, the difference in mean monthly total SWI values between the two scenarios was larger than observed at the mid and high elevations, with



**Figure 6.** The fraction of total mean annual study area surface water input (SWI) generated by each elevation category for each month for control (CTL) and pseudo-global-warming (PGW) scenarios. Elevation categories as follows: high, >3,300 m; mid, 2,300 m < elevation  $\leq$  3,300 m; low,  $\leq$ 2,300 m.

SWI lower in the PGW scenario compared with the CTL scenario. We observed lower SWI in the PGW scenario relative to the CTL scenario in October, March, April, May, July, and August (Table S3 in Supporting Information S1), with the greatest reductions for April and May. Changes in mean daily SWI reinforced the patterns observed at monthly scale. While mid and high elevations showed (a) larger magnitude shifts in late winter and spring SWI (Figure 4) and (b) elevated snowmelt contributions to SWI earlier in the year and reduced contributions later in the year (Figure 5), low elevations displayed more muted changes both to the change in daily SWI as well as the change in daily snowmelt contribution to SWI.

# 3.4.4. Differences Between Control and Pseudo-Global-Warming Scenarios for High and Low Snow Years

Peak SWE and IS in both scenarios differed greatly between 2011 and 2012 (Figure 3), with substantially higher IS, peak SWE and peak SWI during the high snow year (2011) than the low snow year (2012) for CTL and PGW. The timing of these metrics also varied greatly, with peak SWE timing occurring considerably earlier for the low snow year, especially at high elevations. Peak daily SWI for the high snow year for PGW showed greater differences between PGW and CTL values whereas peak SWI was more similar between mean annual and low snow year values for the two scenarios. Peak daily SWI appeared to change the most for PGW high snow years as compared to CTL values. The centroid of SWI (SWI50) for the low snow year for PGW showed much earlier timing as compared to CTL values for the same year.

## 3.4.5. Combined High, Mid and Low Surface Water Input Changes and Basin-Wide Response

Total monthly contributions to study area SWI were lowest for both CTL and PGW during the months of December, January, and February, with spring and summer months exhibiting the highest contributions (Figure 6). However, June contributions for PGW were substantially lower than other spring and summer months, and substantially lower than for CTL, corresponding with lower precipitation for PGW in the preceding month (Figure 2a). Subtracting CTL monthly SWI contributions from PGW monthly contributions revealed higher October to March contributions to total study area SWI for PGW, and lower monthly contributions for June–September for all elevations as compared to CTL. For the months March–May, the direction of monthly contribution changes between PGW and CTL depended on elevation. In March, low elevations showed lower PGW monthly contributions to total annual study area SWI than for CTL, whereas mid and high elevations showed higher values. In April at low and mid elevations, the PGW scenario contributed less to total mean annual study area SWI than the CTL scenario. In May, mid elevations showed substantially lower to total mean annual study area SWI than the CTL, while high elevations had greater PGW monthly contributions to total annual study area SWI than CTL.

Though snowmelt-derived SWI showed similar spatial distributions across the study area for both scenarios (Figures 7a and 7b), less of the total SWI in the PGW scenario came from snowmelt (Figure 7c). Elevation was a major CTL on the percent differences in mean monthly snowmelt-derived SWI between PGW and CTL (Figure 7d, Table S3 in Supporting Information S1), with high elevations showing higher snowmelt-derived SWI

# **Earth's Future**





**Figure 7.** Snowmelt contributions to total water year surface water input (SWI) for control (CTL) (a) pseudo-global-warming (PGW) (b) and PGW minus CTL (c), and the mean monthly PGW minus mean monthly CTL snowmelt contributions to total SWI by elevation range (d). Elevation categories in (d) are as follows: high, >3,300 m; mid, 2,300 m < elevation  $\leq$  3,300 m; low,  $\leq$ 2,300 m. The gray contour in (a,b, and c) is for 2,300 m and separates the low and mid elevation category, while the black contour is for 3,300 m and separates the mid and high elevation category.

values for PGW relative to CTL in fall and spring months, and reductions in snowmelt-derived SWI relative to CTL for summer months. Mid elevations showed higher contributions in fall, winter, and early spring months for PGW as compared to CTL, and reductions during late spring and summer months. Low elevations showed higher values for December and January, declines for October, November, February, March, April, and May, with no change for June to September.

# 4. Discussion

The results of high spatial resolution SnowModel simulations used in this study revealed novel elevational patterns of changes to SWE and SWI under continued global warming. Simulations indicate that high elevation SWE is retained under PGW (Figures 3 and 4, Section 4.1.1). Peak daily SWI was earlier and elevated under PGW, but SWI was also more evenly distributed throughout the year (Figures 3 and 4, Section 4.1.2). PGW led to altered timing, but not complete loss, of snowmelt contributions to SWI (Figures 5–7), with similar annual elevation





Figure 8. Elevation range of present snow telemetry (SNOTEL) snow monitoring stations in comparison to snow water equivalent (SWE) and surface water input (SWI) elevation distributions for control (CTL) and pseudo-global-warming (PGW) scenarios. Q1 is the first quartile of SNOTEL station elevation, Q3 is the third quartile.

contributions of SWI to total study area SWI for CTL and PGW scenarios despite seasonal changes (Figures 6 and 7, Figure S12 in Supporting Information S1, Section 4.1.3). These findings suggest that high resolution snow models that (a) better capture mountain topography (e.g., Figures S17, S18 in Supporting Information S1) and (b) model forcings that better capture convective and orographic processes will lead to different predictions of snow patterns than models without these processes. This is an important consideration for future regional to national scale hydrologic modeling applications in snow dominated areas (Garousi-Nejad & Tarboton, 2021; Sexstone, Driscoll et al., 2020). Additionally, the IS metric (Barnhart et al., 2022) associated with SWI is an important advance in analyzing snowmelt and rainfall inputs to the soil surface. IS captures both timing and intensity of snowmelt rate (Barnhart et al., 2016) fail to fully capture. Finally, we show that existing snow monitoring stations become less representative of future SWE (Figure 8, Section 4.2), with implications for the predictive capacity of future streamflow forecasting.

#### 4.1. Elevation-Dependent SWE and SWI Changes

PGW precipitation was greater for all elevations and all months (0.4–0.9 mm/d), consistent with observed trends for the Northern Front Range of Colorado (Fassnacht et al., 2018, 2020), except for spring, when precipitation declined as compared to CTL (Figure 2). Temperature was higher for all elevations and months (4.9–5.6°C), with the greatest positive changes for late spring and fall months. Precipitation and temperature changes from SnowModel simulations were generally consistent with regional changes from CMIP6 scenarios (Almazroui et al., 2021). The fraction of precipitation falling as snow declined for all elevations and months (up to 40%), except for November–April at high elevations. This decline in snowfall fractions is consistent with results across

the western United States for the mid-21st century (Klos et al., 2014), though our work demonstrates high elevations retain high snowfall fractions in PGW simulations in contrast to prior work showing ubiquitous declines (Rhoades, Jones et al., 2018; Wi et al., 2012).

#### 4.1.1. Retention of High Elevation SWE Under Pseudo-Global-Warming

In response to changes in precipitation and temperature between the two scenarios, simulations revealed peak SWE declined between CTL and PGW for low and mid elevations, on the order of -14% to -45%, while the date of peak SWE uniformly shifted earlier in the year for all elevations. The magnitude of this shift was greatest for mid elevations (~23 days, Figure 3). Declines in peak SWE for PGW in the study area are consistent with (a) Ikeda et al. (2021) who assessed SWE changes between 2070-2099 and 1976-2005 at 4-km resolution and found 30%-60% decreases in April 1 SWE in the region with peak SWE 20-30 days earlier and (b) Siirila-Woodburn et al. (2021) who reported a 25% decline in peak SWE for the western United States for the period 2025–2049, though in our study declines range from -45% for low elevations to -3% for high elevations in the study domain for our period of analysis. Our finding that low elevations show the greatest declines in SWE between CTL and PGW in the study region (Figures 3 and 4) is consistent with historical declines in SWE for similar elevations from 1984 to 2017 in the neighboring Rio Grande Basin (Sexstone, Penn et al., 2020). Limited high elevation snow change for PGW is consistent with Barsugli et al. (2020) who reported limited changes in high elevation April and May SWE using a 250-m resolution model forced with CMIP5 projections for 2041-2070. SnowModel simulations for the Northern Rocky Mountains of Canada similarly project snowpack loss that is strongly elevation and season dependent for the periods 2045-2059 and 2085-2099 as compared to 1979-1994 (Mortezapour et al., 2022). Taken together, these comparisons emphasize that inland, high elevation areas will likely remain conducive to substantial snow accumulation for decades to come, especially when assessing high resolution model output that better resolves mountain topography and snow dynamics. In maritime, snowy regions of North America including the Sierra Nevada and Cascades, high resolution simulations that resolve mountain topography are needed to further understand the degree to which snow loss will occur by elevation.

# 4.1.2. Earlier and Elevated Peak Daily Surface Water Input, yet More Evenly Distributed Input Under Pseudo-Global-Warming

*IS*, which reflects the concentration of input in time, declined between CTL and PGW across all elevations, with similar reductions at all elevations due to higher winter input and lower spring input that result in a more uniform distribution of input throughout the year (Figure 3). Despite these reductions in IS, elevated peak daily SWI for PGW was observed for all elevations. The dates of peak SWI and SWI centroid moved to earlier in the year for all elevations and with the biggest advances for low and mid elevations, consistent with Xu et al. (2021). While prior work has shown earlier streamflow timing in the UCRB for future decades (Wrzesien & Pavelsky, 2020) our work highlights the potential for higher peak flows and a shorter period of elevated snowmelt SWI during snowmelt (Figure 5). Earlier SWI timing for PGW as compared to CTL follow the results from Musselman et al. (2021), who show that snowmelt is already occurring prior to peak SWE in recent years as compared to early SNOTEL records of the 1980s and suggest that these observed patterns will continue in future decades. For low and mid elevations, our results generally align with Kampf and Lefsky (2016) who observed reduced snowmelt contributions to peak flows in areas where peak flows were historically sourced both from rain and snow contributions. Mixed input was a small component of total SWI for both CTL and PGW, consistent with rain on snow analysis by McCabe, Betancourt, and Hidalgo (2007), with mixed input only exceeding 10% of total input.

#### 4.1.3. Altered Timing, but Not Complete Loss, of Snowmelt Contributions to Surface Water Input

In our study, the fraction of total annual SWI generated by snowmelt was lower for all elevations between CTL and PGW (Figures 5 and 7), similar to the results of Li et al. (2017) who showed declines in snowmelt-derived streamflow across the UCRB for RCP4.5 and RCP8.5 out to 2,100. Seasonal snowmelt contributions to SWI were higher for fall, winter, and spring months at mid elevations (Figure 5, Figure 7d), but lower for summer months in PGW, differing from the results of Li et al. (2017) who showed reductions for snowmelt generated streamflow for all months. Shifts in the timing of total SWI during spring months was dependent on elevation (Figure 6), with low and mid elevations displaying these shifts earlier in the year than high elevations. Half of the SWI for in the study area CTL and for PGW was generated by elevations >2,700 m (only  $\sim$ 37% of study area, Figure 1), generally consistent with prior research in the UCRB and the neighboring Cache la Poudre watershed where half of the

streamflow is generated from elevations >3,000 m (Hammond et al., 2018; Harrison et al., 2021). Cumulative contributions to total study area SWI at water year scale retain a nearly identical relation with elevation for PGW as compared to CTL (Figure S12 in Supporting Information S1) despite the shifts in timing of sub-annual SWI.

#### 4.1.4. Implications for the Future of Timing and Magnitude of Water Availability

The combined, and potentially countervailing, effects of greater precipitation and earlier, flashier SWI complicate predictions for the future timing and magnitude of streamflow. The potential for greater precipitation in future decades, similar to what is shown by the PGW scenario, may offset streamflow reductions caused by the loss of efficiently generated snowmelt from seasonal snowpacks at mid elevations (Reclamation, 2016). However, future precipitation changes are uncertain (Almazroui et al., 2021), and while simulations suggest future droughts in western North America will be more spatially pervasive and extreme, uncertainties in vegetation processes and precipitation representation in models complicate the interpretation of these results (Cook et al., 2020). Additionally, warming during spring and summer months could partition summer SWI to soil and vegetation evaporative losses rather than streamflow or groundwater recharge (Barnhart et al., 2020; Gordon et al., 2022; Hammond et al., 2019; Harpold & Molotch, 2015; Tague & Peng, 2013), as earlier snow disappearance can lead to earlier ET fluxes.

Prior work has shown that snowmelt rate and timing are linked (Alonso-González et al., 2022; Serreze et al., 1999), with slower snowmelt associated with decreased streamflow generation (Barnhart et al., 2016; Musselman, Molotch, & Molotch, 2017). However, recent research also highlights that earlier snowmelt timing and associated higher runoff efficiency (runoff generated per unit of input) may compensate for lower future snowmelt rates (Barnhart et al., 2020; Gordon et al., 2022; Hammond & Kampf, 2020; Jeton et al., 1996). The competition of these two mechanisms may dictate how runoff from snow-dominated sites might change in the future as snowmelt rate and timing change under future climate conditions. Barnhart et al. (2020) report that snowmelt season runoff was governed by the competing influence of snowmelt timing and snowmelt rate via a modeling study where each factor was considered separately due to the linkage between snowmelt rate and timing (Trujillo et al., 2014). The simulated changes in subsurface storage and melt season runoff were more sensitive to changes in snowmelt timing than snowmelt rate for a site in Colorado, suggesting greater runoff from earlier snowmelt may counteract runoff losses due to slower snowmelt and that this process is mediated by plant available water storage and energy availability. In the Lower Colorado River Basin, the effects of temperature-induced snow loss on annual streamflow appear to have been moderated by enhanced winter hydrological inputs and streamflow production (Robles et al., 2021).

In the current hydroclimate, streamflow generation for many western U.S. mountain watersheds tends to peak in spring whereas runoff efficiency peaks in winter. As watersheds become warm enough to experience winter melt and rain on snow, this enhanced winter runoff efficiency may act as a temporary buffer against reductions to streamflow generated by concentrated spring and summer snowmelt (Hammond et al., 2020). Results of the present study indicate earlier SWI with higher peak SWI values at all elevations (Figure 3). This result, which integrates changes in snowmelt and precipitation phase through SWI, contrasts with Musselman, Clark, Liu et al. (2017), Musselman, Molotch and Margulis (2017), who, using the same meteorological forcing data as this study, reported declines in future snowmelt input to the hydrologic system and possible concomitant declines in streamflow. Earlier SWI may lead to efficient streamflow generation if soil moisture is conducive to infiltration excess overland flow, saturation excess overland flow, or shallow subsurface flow. However, for warmer low and mid elevations where snowmelt is a smaller component of SWI, earlier SWI was simulated to be associated with reduced annual total SWI (Hale et al., 2022). Integration of high resolution SnowModel output with a hydrological model would be useful to assess study-area-wide hydrologic partitioning and streamflow generation for future periods. This integration is needed to examine the combined effects of (a) changes occurring to deep snowpacks in relatively small areas of the western United States and (b) the near loss of shallow snowpacks across large areas on basin-wide water availability.

The combined effects of shifts in timing and magnitude in SWE and SWI also have broad implications for reservoir management and wildfire dynamics. Continued warming is diminishing role of snowy mountains as natural reservoirs, and thus water resource management will likely need to adjust to shifting reservoir inflow timing and magnitude (Rhoades, Ullrich et al., 2018). With drought severity and extent expected to increase in coming decades (Cook et al., 2020; Gutzler & Robbins, 2011), interannual variability in annual SWE and SWI metrics (highlighted by years 2011 and 2012, Figure 3) could test coordinated reservoir management for

multiple objectives. Even if annual streamflow remains the same or increases slightly, it could be hard for coordinated reservoir management to adjust to changes in the timing of inflows (Cohen et al., 2020); Li et al., 2010; Sterle et al., 2020), given a greater offset between water supply and demand timing. These changes will be most impactful and relevant for water management decision making when hydroclimatic shifts for individual years compound and result in changes to decadal and multi-decadal variability (Garrick et al., 2008; Hurkmans et al., 2009; McCabe, Clark, & Hay, 2007; Nowak et al., 2012; NRC, 2007).

Hedging in reservoirs, the practice of storing more water during the spring to meet the widening gap between inflows and outflow demand during the summer low flow period, may be needed to respond to shifting seasonal reservoir inflows (Jones & Hammond, 2020). Other potential adaptations to shifting SWI timing and magnitude for downstream water availability include the construction of new surface water reservoirs or the enhancement of existing infrastructure (Perry et al., 2017; Reclamation, 2021), the adjustment of current reservoir management strategies (Delaney et al., 2020; Reclamation, 2021; Sumargo et al., 2020), and recharging groundwater during periods of excess (Scanlon et al., 2016; Zhang, 2015), and according to Kellner and Brunner (2021) immediate action is required to plan for balanced upstream-downstream water availability in water infrastructure and management decision making. Over the past several decades, earlier snowmelt occurrence and reduced peak SWE have corresponded with increasing solar forcing on snow in the western U.S (Gleason et al., 2016; Westerling et al., 2006) as well as with increasing solar forcing on snow in the western U.S (Gleason et al., 2019). Wildfires have increasingly occurred in seasonally snow covered areas from 1984 to 2020 throughout the western United States (Kampf et al., 2022). Continued snow loss and earlier melt may lead to larger and more intense fires in these areas in future decades, though potential increases in spring and summer precipitation could have a countervailing influence.

#### 4.2. Snow Measurement Network Coverage and the Distributions of SWE and SWI

Current snowpack monitoring does not represent the range of land covers, elevations and terrain classes found in mountainous regions (Fassnacht et al., 2012; Kampf et al., 2020; Sexstone & Fassnacht, 2014), partially due to the difficulties of measuring snow in alpine areas. Stations do not always reflect their nearby surroundings (Meromy et al., 2013), and conditions in mountainous regions experience substantial inter-annual variability (Fassnacht et al., 2003, 2012) especially in the alpine (Sexstone et al., 2018). The 75 existing SNOTEL sites in the study region range in elevation from 2,605 to 3,532 m, with a mean elevation of 3,094 and median of 3,127 m. Although the mean and median elevation of the SNOTEL sites is greater than the median and median elevations of the modeling domain, the SNOTEL site elevation range does not cover much of the persistent snow zone above 3,500 m in the high elevation category (Figure 8). Prior work has suggested these networks will have reduced predictive capacity for streamflow forecasting as snowpacks continue to decline (Livneh & Badger, 2020). A lack of observations at both the high and low extents of where snow accumulates could inhibit the ability to predict snow and streamflow conditions.

For both CTL and PGW, the existing network of SNOTEL stations does not monitor areas with the highest mean annual peak SWE values (Figure 8a), though it does sample the elevations with the greatest volumetric SWE in both scenarios (Figure 8b). With the greatest declines in SWE for low and mid elevations for PGW, greater future volumetric SWE will occur at high elevations not sampled by the existing network. Adding sites at higher elevations not presently monitored could allow for better tracking the remaining water content within snow to provide information for downstream streamflow forecasting and coordination. However, in future decades, treeline may advance in elevation, shrinking alpine areas (Kummel et al., 2021), potentially making existing stations more reflective of study area snowpacks in the future sub-alpine. The study area examined here is generally higher in elevation and snowier than other areas of the UCRB (Figures 1a and 1b), so additional snow monitoring in this area may provide more useful information on the distribution of SWE and snowmelt-derived SWI that drives water availability than instrumentation in other areas of the basin.

Considering the volumetric distribution of mean annual SWI (Figure 8c), the existing SNOTEL network generally samples elevations above those with greatest SWI. Adding additional sites to slightly lower elevations could better capture the timing and rate of rainfall and snowmelt input both in the near-term and long-term. Low and mid elevation sites would also better capture mid-winter melt not currently measured by exiting monitoring stations in low and mid elevations that measure precipitation but not SWE or soil moisture. Given that a greater fraction of precipitation in future years will fall as rain rather than snow (Figure 2, Klos et al., 2014), precipitation measurements at low and mid elevation sites may suffer less from under catch since snow is more susceptible to this phenomenon than rain (Groisman & Easterling, 1994). USGS gaged headwater watersheds are generally more uniformly distributed across elevations than SNOTEL sites, and although there are a number of gaged headwater watersheds draining only the high elevation persistent snow zone, there are fewer gaged watersheds isolating the transitional or low snow zones; rather, gages at lower elevations tend to integrate the hydrologic response from multiple snow zones (Kampf et al., 2020). Thus, additional gaging better isolating the hydrologic response to different snow zones could also provide information for downstream flow forecasting and coordination. Additionally, soil moisture is sparsely monitored in the basin, but additional soil moisture monitoring along elevation gradients in the basin may help to mitigate lessened predictive capacity due to declining snowpacks at existing snow monitoring stations (Livneh & Badger, 2020).

#### 4.3. Limitations, Uncertainties and Areas for Continued Research

Though the results of this study were coherent between analyses implemented at multiple time scales (daily, monthly, annually) and broadly consistent with prior research, several factors limit the scope of inference of our study. First, our model evaluation revealed poorer performance in low snow areas, areas with limited existing snow monitoring to use for model calibration and validation. Further work is needed to evaluate whether modeling snow accurately in these low snow areas is important for assessing water availability at management relevant spatial and temporal scales. Second, though SNOTEL sites use Alter wind shields to increase precipitation catch efficiency (Serreze et al., 1999), under catch has been approximated at ~11% (Yang et al., 1998), with up to 30% in an intense snowstorm with high winds in Colorado (Rasmussen, Liu et al., 2011). Our use of SNOTEL data for model evaluation is thus susceptible to this source of error (Scalzitti et al., 2016a, 2016b). Additionally, several of our analyses were based on aggregation of model output to elevation bands, with local patterns lost in the process. Further analysis of model output for sub-regions may identify nuanced patterns compared to those identified using study-area-wide elevational aggregation.

There are also limitations associated with the regional climate model data used as input to this study's SnowModel simulations. Liu et al. (2017) broadly characterized the biases inherent in the WRF PGW data set and noted a warm and dry bias in the central U.S. in the late summer months. However, given that this issue is related to convective storms in the late summer, it is unlikely that this would substantially affect the main results of our analyses. There are also limitations to using a single model representation of future climate precipitation given that it is not possible to characterize uncertainty with this method. As ensemble convection-permitting simulations become available on longer timescales, future research could analyze changes in snowpack from an ensemble framework to assess uncertainty. Finally, although the PGW methodology does not represent large-scale changes in storm tracks associated with climate change, the methodology allows assessment of specific time periods and see how actual weather events in that time period might have been different if they experienced thermodynamic climate warming and enhanced moisture.

Subsequent snow modeling efforts could explore including dynamical vegetation to account for increased forest cover at high elevations (e.g., Barnhart et al., 2021), to explore the effects of significantly changed canopy structure due to more prevalent wildfire in snowy areas (Kampf et al., 2022), and to address the effects of forest disturbance on snowpack and streamflow (Goeking & Tarboton, 2020; Varhola et al., 2010). The existing simulations might also be used to examine differences in SP by aspect and landcover (e.g., Barsugli et al., 2020; Sexstone, Penn et al., 2020). While SWI as used in this analysis provides an initial estimate of runoff resulting from rainfall and snowmelt input available at the soil surface, this estimate does not account for the various pathways that SWI could take into the soil, across the subsurface, or into the atmosphere. Thus, connecting the surface water inputs from this study to a hydrological model would enable detailed exploration of changes to evapotranspiration, soil moisture, groundwater recharge, and streamflow generation due to snow loss, changes in SWI timing and magnitude, and changes to the synchrony of energy and water availability (e.g., Gordon et al., 2022).

# 5. Conclusion

The results of high spatial resolution PGW SnowModel simulations for the Colorado portion of the UCRB revealed widespread SWE loss, with reductions in peak SWE at low (-45%, -30 mm) and mid (-14%, -26 mm) elevations. However, high elevation, continental areas in the bounds of this study, and elsewhere in the UCRB,



are projected to experience a more limited peak SWE change (-3%) and will likely remain conducive to seasonal snow accumulation and melt for decades to come. Peak SWE and peak SWI occurred earlier in the year (peak SWE, 17-23 days; peak SWI, 14-23 days) for all elevations under PGW, with higher peak daily SWI magnitude (30%-42%, 10-12 mm/d) also noted at all elevations. Despite earlier and elevated peak daily SWI, reduced IS revealed more evenly distributed SWI under PGW. Snowmelt-derived surface water inputs were lower at the annual scale, but experienced seasonal growth and shifts in timing dependent on elevation for PGW as compared to CTL, with low elevations experiencing shifts earlier in the year. Finally, existing snow monitoring stations became less representative of future SWE, because there are few monitoring sites at high elevations where future SWE will contribute a larger fraction of total SWI compared to current climatic conditions. Water stored in snowpacks has historically served as a reliable extension of manmade reservoir storage, providing sustained streamflow generation throughout spring and summer months. The combined effects of shifts in timing and magnitude in SWE and SWI have broad implications for (a) reservoir management given a greater offset between water supply and demand timing under PGW, and (b) future wildfire dynamics because large fires are already occurring in snowy mountain areas where SWE has declined, and snowmelt has shifted earlier in the year. These results and added insights suggest high resolution snow models that better capture mountain topography and model forcings that better capture convective and orographic processes may improve representation of seasonal snow accumulation and melt dynamics, elevational variability in snow patterns, and changes in precipitation phase under future climate conditions. This is an important consideration for future regional to national scale hydrologic modeling applications in snow dominated areas. Additional work that connects high resolution SnowModel outputs to a hydrological model could benefit downstream coordinated water management by to evaluating the combined effects of widespread snow loss and earlier, flashier SWI.

## **Data Availability Statement**

The data that support the findings of this study, including model input and output, are available in USGS Science-Base at https://doi.org/10.5066/P9K0QUCN.

## References

- Adler, R. W. (2008). Revisiting the Colorado River compact: Time for a change. Journal of Land Resources Environmental Law, 28, 19.
- Almazroui, M., Islam, M. N., Saeed, F., Saeed, S., Ismail, M., Ehsan, M. A., et al. (2021). Projected changes in temperature and precipitation over the United States, Central America, and the Caribbean in CMIP6 GCMs. *Earth Systems and Environment*, 5(1), 1–24. https://doi.org/1 0.1007/s41748-021-00199-5
- Alonso-González, E., Revuelto, J., Fassnacht, S. R., & López-Moreno, J. I. (2022). Combined influence of maximum accumulation and melt rates on the duration of the seasonal snowpack over temperate mountains. *Journal of Hydrology*, 608, 127574. https://doi.org/10.1016/j. jhydrol.2022.127574
- Ban, Z., & Lettenmaier, D. P. (2022). Asymmetry of Western US river basin sensitivity to seasonally varying climate warming. Water Resources Research, 58(2), e2021WR030367. https://doi.org/10.1029/2021wr030367
- Barnhart, T. B., Farmer, W. H., Hammond, J. C., Sexstone, G. A., Curran, J. H., Koch, J. C., & Driscoll, J. M. (2022). Evaluating hydrologic region assignment techniques for ungaged basins in Alaska, USA. *River Research and Applications*, 38(9), 1569–1584. https://doi. org/10.1002/tra.4028
- Barnhart, T. B., Molotch, N. P., Livneh, B., Harpold, A. A., Knowles, J. F., & Schneider, D. (2016). Snowmelt rate dictates streamflow. Geophysical Research Letters, 43(15), 8006–8016. https://doi.org/10.1002/2016gl069690
- Barnhart, T. B., Tague, C. L., & Molotch, N. P. (2020). The counteracting effects of snowmelt rate and timing on runoff. Water Resources Research, 56(8), e2019WR026634. https://doi.org/10.1029/2019wr026634
- Barnhart, T. B., Vukomanovic, J., Bourgeron, P., & Molotch, N. P. (2021). Future land cover and climate may drive decreases in snow wind-scour and transpiration, increasing streamflow at a Colorado, USA headwater catchment. *Hydrological Processes*, 35(11), e14416. https://doi.org/ 10.1002/hyp.14416
- Barsugli, J. J., Ray, A. J., Livneh, B., Dewes, C. F., Heldmyer, A., Rangwala, I., et al. (2020). Projections of mountain snowpack loss for wolverine denning elevations in the Rocky Mountains. *Earth's Future*, 8(10), e2020EF001537. https://doi.org/10.1029/2020ef001537
- Baston, D. (2020). Exactextractr: Fast extraction from Raster Datasets using Polygons. R package version 0.2.0. Retrieved from https://CRAN. R-project.org/package=exactextractr
- Berghuijs, W. R., Woods, R. A., & Hrachowitz, M. (2014). A precipitation shift from snow towards rain leads to a decrease in streamflow. Nature Climate Change, 4(7), 583–586. https://doi.org/10.1038/nclimate2246
- Butler, A., Jerla, C., Nowak, K., Prairie, J., Oakley, B., Wilson, N., & Zagona, E. (2015). The Colorado River Basin water supply and demand study: Modeling to support a robust planning framework. In *Proceedings of the 3rd Joint Federal Interagency Conference on Sedimentation* and Hydrologic Modeling (pp. 19–21).
- Canada Centre for Remote Sensing. (2020). Canada Centre for Mapping and Earth Observation, Natural Resources Canada, Comisión Nacional para el Conocimiento y Uso de la Biodiversidad, Comisión Nacional Forestal, Instituto Nacional de Estadística y Geografía, U.S. Geological Survey, 2020: 2010 Land Cover of North America at 30 meters. Version 2.0 Retrieved from http://www.cec.org/tools-and-resources/map-files/land-cover-2010-landsat-30m
- Capesius, J. P., & Stephens, V. C. (2009). Regional regression equations for estimation of natural streamflow statistics in Colorado. Scientific Investigations Report 2009–5136. U.S. Geological Survey, 5136, 46.

#### Acknowledgments

This research was funded by the U.S. Geological Survey (USGS) Water Availability and Use Science Program as part of the Water Resources Mission Area Snow Hydrology Research Project. Additional funding support was provided by the Colorado Water Center at Colorado State University. Computing resources were provided by USGS Core Science Systems Advanced Research Computing Center. Thanks to Gregory McCabe (USGS, Water Mission Area) and two anonymous reviewers for their suggestions and edits that improved this article. Any use of trade, firm, or product names is for descriptive purposes only and does not imply endorsement by the U.S. Government.

- Christensen, N. S., & Lettenmaier, D. P. (2007). A multimodel ensemble approach to assessment of climate change impacts on the hydrology and water resources of the Colorado River Basin. *Hydrology and Earth System Sciences*, 11(4), 1417–1434. https://doi.org/10.5194/ hess-11-1417-2007
- Christensen, N. S., Wood, A. W., Voisin, N., Lettenmaier, D., & Palmer, R. N. (2004). The effects of climate change on the hydrology and water resources of the Colorado River Basin. *Climate Change*, 62(1–3), 337–363. https://doi.org/10.1023/b:clim.0000013684.13621.1f
- Cohen, J., Ye, H., & Jones, J. (2015). Trends and variability in rain-on-snow events. *Geophysical Research Letters*, 42(17), 7115–7122. https://doi.org/10.1002/2015GL065320
- Cohen, J. S., Zeff, H. B., & Herman, J. D. (2020). Adaptation of multiobjective reservoir operations to snowpack decline in the Western United States. *Journal of Water Resources Planning and Management*, 146(12). https://doi.org/10.1061/(asce)wr.1943-5452.0001300
- Cook, B. I., Mankin, J. S., Marvel, K., Williams, A. P., Smerdon, J. E., & Anchukaitis, K. J. (2020). Twenty-first century drought projections in the CMIP6 forcing scenarios. *Earth's Future*, 8(6), e2019EF001461. https://doi.org/10.1029/2019ef001461
- Dai, A. (2008). Temperature and pressure dependence of the rain-snow phase transition over land and ocean. *Geophysical Research Letters*, 35(12), L12802. https://doi.org/10.1029/2008GL033295
- Delaney, C. J., Hartman, R. K., Mendoza, J., Dettinger, M., Delle Monache, L., Jasperse, J., et al. (2020). Forecast informed reservoir operations using ensemble streamflow predictions for a multipurpose reservoir in Northern California. *Water Resources Research*, 56(9), e2019WR026604. https://doi.org/10.1029/2019wr026604
- Dingman, S. L. (2002). Physical Hydrology (2nd ed.). 646. Prentice Hall.
- Dudley, R. W., Hodgkins, G. A., McHale, M. R., Kolian, M. J., & Renard, B. (2017). Trends in snowmelt-related streamflow timing in the conterminous United States. *Journal of Hydrology*, 547, 208–221. https://doi.org/10.1016/j.jhydrol.2017.01.051
- Eurich, A., Kampf, S. K., Hammond, J. C., Ross, M., Willi, K., Vorster, A. G., & Pulver, B. (2021). Predicting mean annual and mean monthly streamflow in Colorado ungauged basins. *River Research and Applications*, 37(4), 569–578. https://doi.org/10.1002/tra.3778
- Fassnacht, S. R., Dressler, K. A., & Bales, R. C. (2003). Snow water equivalent interpolation for the Colorado River Basin from snow telemetry (SNOTEL) data. Water Resources Research, 39(8), 1208. https://doi.org/10.1029/2002wr001512
- Fassnacht, S. R., Dressler, K. A., Hultstrand, D. M., Bales, R. C., & Patterson, G. (2012). Temporal inconsistencies in coarse-scale snow water equivalent patterns: Colorado River Basin snow telemetry-topography regressions. *Pirineos*, 167(167), 165–185. https://doi.org/10.3989/ pirineos.2012.167008
- Fassnacht, S. R., Patterson, G. G., Venable, N. B., Cherry, M. L., Pfohl, A. K., Sanow, J. E., & Tedesche, M. E. (2020). How do we define climate change? Considering the temporal resolution of Niveo-Meteorological data. *Hydrology*, 7(3), 38. https://doi.org/10.3390/hydrology7030038
- Fassnacht, S. R., Venable, N. B., McGrath, D., & Patterson, G. G. (2018). Sub-seasonal snowpack trends in the rocky mountain national park area, Colorado, USA. Water, 10(5), 562. https://doi.org/10.3390/w10050562
- Fyfe, J. C., Derksen, C., Mudryk, L., Flato, G. M., Santer, B. D., Swart, N. C., et al. (2017). Large near-term projected snowpack loss over the Western United States. *Nature Communications*, 8(1), 1–7. https://doi.org/10.1038/ncomms14996
- Garousi-Nejad, I., & Tarboton, D. G. (2021). A comparison of national water model retrospective analysis snow outputs at SNOTEL sites across the Western US. Hydrological Processes, e14469.
- Garrick, D., Jacobs, K., & Garfin, G. (2008). Models, assumptions, and stakeholders: Planning for water supply variability in the Colorado River Basin 1. JAWRA Journal of the American Water Resources Association, 44(2), 381–398. https://doi.org/10.1111/j.1752-1688.2007.00154.x
- GDAL/OGR contributors. (2022). GDAL/OGR geospatial data abstraction software library. Open Source Geospatial Foundation. https://doi. org/10.5281/zenodo.5884351
- Gleason, K. E., McConnell, J. R., Arienzo, M. M., Chellman, N., & Calvin, W. M. (2019). Four-fold increase in solar forcing on snow in Western US burned forests since 1999. *Nature Communications*, 10(1), 1–8. https://doi.org/10.1038/s41467-019-09935-y
- Goeking, S. A., & Tarboton, D. G. (2020). Forests and water yield: A synthesis of disturbance effects on streamflow and snowpack in Western coniferous forests. *Journal of Forestry*, 118(2), 172–192. https://doi.org/10.1093/jofore/fvz069
- Gordon, B. L., Brooks, P. D., Krogh, S. A., Boisrame, G. F., Carroll, R. W., McNamara, J. P., & Harpold, A. A. (2022). Why does snowmelt-driven streamflow response to warming vary? A data-driven review and predictive framework. *Environmental Research Letters*, 17(5), 053004. https://doi.org/10.1088/1748-9326/ac64b4
- Greene, E. M., Liston, G. E., & Pielke, R. A. (1999). Simulation of above treeline snowdrift formation using a numerical snow-transport model. Cold Regions Science and Technology, 30(1–3), 135–144. https://doi.org/10.1016/S0165-232X(99)00008-7
- Groisman, P. Y., & Easterling, D. R. (1994). Variability and trends of total precipitation and snowfall over the United States and Canada. Journal of Climate, 7(1), 184–205. https://doi.org/10.1175/1520-0442(1994)007<0184:VATOTP>2.0.CO;2
- Guan, B., Waliser, D. E., Ralph, F. M., Fetzer, E. J., & Neiman, P. J. (2016). Hydrometeorological characteristics of rain-on-snow events associated with atmospheric rivers. *Geophysical Research Letters*, 43(6), 2964–2973. https://doi.org/10.1002/2016gl067978
- Gutzler, D. S., & Robbins, T. O. (2011). Climate variability and projected change in the Western United States: Regional downscaling and drought statistics. *Climate Dynamics*, 37(5), 835–849. https://doi.org/10.1007/s00382-010-0838-7
- Hale, K. E., Wlostowski, A. N., Badger, A. M., Musselman, K. N., Livneh, B., & Molotch, N. P. (2022). Modeling streamflow sensitivity to climate warming and surface water inputs in a montane catchment. *Journal of Hydrology: Regional Studies*, 39, 100976. https://doi.org/10.1016/j.ejrh.2021.100976
- Hammond, J. C. (2020). Contiguous U.S. annual snow persistence and trends from 2001-2020. U.S. Geological Survey data release. https:// doi.org/10.5066/P9U7U5FP
- Hammond, J. C., Harpold, A. A., Weiss, S., & Kampf, S. K. (2019). Partitioning snowmelt and rainfall in the critical zone: Effects of climate type and soil properties. *Hydrology and Earth System Sciences*, 23(9), 3553–3570. https://doi.org/10.5194/hess-23-3553-2019
- Hammond, J. C., & Kampf, S. K. (2020). Subannual streamflow responses to rainfall and snowmelt inputs in snow-dominated watersheds of the Western United States. *Water Resources Research*, 56(4), e2019WR026132. https://doi.org/10.1029/2019wr026132
- Hammond, J. C., Saavedra, F. A., & Kampf, S. K. (2018). How does snow persistence relate to annual streamflow in mountain watersheds of the Western US with wet maritime and dry continental climates? *Water Resources Research*, 54(4), 2605–2623. https://doi.org/10.1002/ 2017wr021899
- Harpold, A. A., & Molotch, N. P. (2015). Sensitivity of soil water availability to changing snowmelt timing in the Western US. *Geophysical Research Letters*, 42(19), 8011–8020. https://doi.org/10.1002/2015gl065855
- Harrison, H. N., Hammond, J. C., Kampf, S., & Kiewiet, L. (2021). On the hydrological difference between catchments above and below the intermittent-persistent snow transition. *Hydrological Processes*, 35(11), e14411. https://doi.org/10.1002/hyp.14411
- Hiemstra, C. A., Liston, G. E., & Reiners, W. A. (2006). Observing, modelling, and validating snow redistribution by wind in a Wyoming upper treeline landscape. *Ecological Modelling*, 197(1–2), 35–51. https://doi.org/10.1016/j.ecolmodel.2006.03.005

- Hurkmans, R., Troch, P. A., Uijlenhoet, R., Torfs, P., & Durcik, M. (2009). Effects of climate variability on water storage in the Colorado River Basin. Journal of Hydrometeorology, 10(5), 1257–1270. https://doi.org/10.1175/2009jhm1133.1
- Huss, M., Bookhagen, B., Huggel, C., Jacobsen, D., Bradley, R. S., Clague, J. J., et al. (2017). Toward mountains without permanent snow and ice. *Earth's Future*, 5(5), 418–435. https://doi.org/10.1002/2016ef000514
- Ikeda, K., Rasmussen, R., Liu, C., Newman, A., Chen, F., Barlage, M., et al. (2021). Snowfall and snowpack in the Western US as captured by convection permitting climate simulations: Current climate and pseudo global warming future climate. *Climate Dynamics*, 57(7–8), 1–25. https://doi.org/10.1007/s00382-021-05805-w
- Jeton, A. E., Dettinger, M. D., & Smith, J. L. (1996). Potential effects of climate change on streamflow, eastern and western slopes of the Sierra Nevada (Vol. 95, p. 4260). US Department of the Interior, US Geological Survey.
- Jones, J. A., & Hammond, J. C. (2020). River management response to multi-decade changes in timing of reservoir inflows, Columbia River Basin, USA. *Hydrological Processes*, 34(25), 4814–4830. https://doi.org/10.1002/hyp.13910
- Kampf, S. K., Burges, S. J., Hammond, J. C., Bhaskar, A., Covino, T. P., Eurich, A., et al. (2020). The case for an open water balance: Re-envisioning network design and data analysis for a complex, uncertain world. Water Resources Research, 56(6), e2019WR026699. https://doi.org/10.1029/2019wr026699
- Kampf, S. K., & Lefsky, M. A. (2016). Transition of dominant peak flow source from snowmelt to rainfall along the Colorado Front Range: Historical patterns, trends, and lessons from the 2013 Colorado Front Range floods. Water Resources Research, 52(1), 407–422. https://doi. org/10.1002/2015wr017784
- Kampf, S. K., McGrath, D., Sears, M. G., Fassnacht, S. R., Kiewiet, L., & Hammond, J. C. (2022). Increasing wildfire impacts on snowpack in the Western US. Proceedings of the National Academy of Sciences United States of America, 119(39), e2200333119. https://doi.org/10.1073/pn as.2200333119
- Kellner, E., & Brunner, M. I. (2021). Reservoir Governance in world's water towers needs to anticipate multi-purpose use. *Earth's Future*, 9(1), e2020EF001643. https://doi.org/10.1029/2020ef001643
- Kiewiet, L., Trujillo, E., Hedrick, A., Havens, S., Hale, K., Seyfried, M., et al. (2022). Effects of spatial and temporal variability in surface water inputs on streamflow generation and cessation in the rain–snow transition zone. *Hydrology and Earth System Sciences*, 26(10), 2779–2796. https://doi.org/10.5194/hess-26-2779-2022
- Kim, R. S., Kumar, S., Vuyovich, C., Houser, P., Lundquist, J., Mudryk, L., et al. (2021). Snow Ensemble Uncertainty Project (SEUP): Quantification of snow water equivalent uncertainty across North America via ensemble land surface modeling. *The Cryosphere*, 15(2), 771–791. https://doi.org/10.5194/tc-15-771-2021
- Klos, P. Z., Link, T. E., & Abatzoglou, J. T. (2014). Extent of the rain-snow transition zone in the Western U.S. under historic and projected climate. *Geophysical Research Letters*, 41(13), 4560–4568. https://doi.org/10.1002/2014GL060500
- Kormos, P. R., Marks, D., McNamara, J. P., Marshall, H. P., Winstral, A., & Flores, A. N. (2014). Snow distribution, melt and surface water inputs to the soil in the mountain rain–snow transition zone. *Journal of Hydrology*, 519, 190–204. https://doi.org/10.1016/j.jhydrol.2014.06.051
- Kummel, M., McGarigal, A., Kummel, M., Earnest, C., & Feiden, M. (2021). Tree establishment and growth drive treeline advance and change treeline form on Pikes Peak (Colorado) in response to recent anthropogenic warming. *Canadian Journal of Forest Research*, 51(10), 1458– 1471. https://doi.org/10.1139/cjfr-2019-0358
- Li, D., Wrzesien, M. L., Durand, M., Adam, J., & Lettenmaier, D. P. (2017). How much runoff originates as snow in the Western United States, and how will that change in the future? *Geophysical Research Letters*, 44(12), 6163–6172. https://doi.org/10.1002/2017gl073551
- Li, L., Xu, H., Chen, X., & Simonovic, S. P. (2010). Streamflow forecast and reservoir operation performance assessment under climate change. Water Resources Management, 24(1), 83–104. https://doi.org/10.1007/s11269-009-9438-x
- Liston, G. E. (1995). Local advection of momentum, heat, and moisture during the melt of patchy snow covers. *Journal of Applied Meteorology*, 34(7), 1705–1715. https://doi.org/10.1175/1520-0450-34.7.1705
- Liston, G. E. (2004). Representing subgrid snow cover heterogeneities in regional and global models. *Journal of Climate*, 17(6), 1381–1397. https://doi.org/10.1175/1520-0442(2004)017<1381:rsschi>2.0.co;2
- Liston, G. E., & Elder, K. (2006a). A distributed snow-evolution modeling system (SnowModel). Journal of Hydrometeorology, 7(6), 1259–1276. https://doi.org/10.1175/JHM548.1
- Liston, G. E., & Elder, K. (2006b). A meteorological distribution system for high-resolution terrestrial modeling (MicroMet). Journal of Hydrometeorology, 7(2), 217–234. https://doi.org/10.1175/JHM486.1
- Liston, G. E., & Hall, D. K. (1995). An energy balance model of lake ice evolution. Journal of Glaciology, 41(138), 373–382. https://doi. org/10.1017/S0022143000016245
- Liston, G. E., & Hiemstra, C. A. (2008). A simple data assimilation system for complex snow distributions (SnowAssim). Journal of Hydrometeorology, 9(5), 989–1004. https://doi.org/10.1175/2008JHM871.1
- Liston, G. E., Itkin, P., Stroeve, J., Tschudi, M., Stewart, J. S., Pedersen, S. H., et al. (2020). A Lagrangian snow-evolution system for sea-ice applications (SnowModel-LG): Part I—Model description. *Journal of Geophysical Research: Oceans*, 125(10), e2019JC015913. https://doi. org/10.1029/2019JC015913
- Liston, G. E., & Sturm, M. (1998). A snow-transport model for complex terrain. Journal of Glaciology, 44(148), 498–516. https://doi.org/10.1 017/S0022143000002021
- Liu, C., Ikeda, K., Rasmussen, R., Barlage, M., Newman, A. J., Prein, A. F., et al. (2017). Continental-scale convection-permitting modeling of the current and future climate of North America. *Climate Dynamics*, 49(1), 71–95. https://doi.org/10.1007/s00382-016-3327-9
- Livneh, B., & Badger, A. M. (2020). Drought less predictable under declining future snowpack. *Nature Climate Change*, 10(5), 452–458. http s://doi.org/10.1038/s41558-020-0754-8
- López-Moreno, J. I., Fassnacht, S. R., Heath, J. T., Musselman, K. N., Revuelto, J., Latron, J., et al. (2013). Small scale spatial variability of snow density and depth over complex alpine terrain: Implications for estimating snow water equivalent. Advances in Water Resources, 55, 40–52. https://doi.org/10.1016/j.advwatres.2012.08.010
- Lukas, J., & Payton, E. (Eds.) (2020). Colorado River Basin climate and Hydrology: State of the science. (p. 531). Western Water Assessment, University of Colorado Boulder.
- Lundquist, J. D., Neiman, P. J., Martner, B., White, A. B., Gottas, D. J., & Ralph, F. M. (2008). Rain versus snow in the Sierra Nevada, California: Comparing Doppler profiling radar and surface observations of melting level. *Journal of Hydrometeorology*, 9(2), 194–211. https://doi. org/10.1175/2007JHM853.1
- Markham, C. G. (1970). Seasonality of precipitation in the United States. Annals of the Association of American Geographers, 60(3), 593–597. https://doi.org/10.1111/j.1467-8306.1970.tb00743.x

- McCabe, G. J., Betancourt, J. L., & Hidalgo, H. G. (2007). Associations of decadal to multidecadal sea-surface temperature variability with upper Colorado River Flow 1. JAWRA Journal of the American Water Resources Association, 43(1), 183–192. https://doi.org/10.1111/j.1 752-1688.2007.00015.x
- McCabe, G. J., Clark, M. P., & Hay, L. E. (2007). Rain-on-snow events in the Western United States. Bulletin of the American Meteorological Society, 88(3), 319–328. https://doi.org/10.1175/bams-88-3-319
- McCabe, G. J., & Wolock, D. M. (2007). Warming may create substantial water supply shortages in the Colorado River Basin. Geophysical Research Letters, 34(22), L22708. https://doi.org/10.1029/2007GL031764
- McCabe, G. J., & Wolock, D. M. (2011). Independent effects of temperature and precipitation on modeled runoff in the conterminous United States. Water Resources Research, 47(11), W11522. https://doi.org/10.1029/2011wr010630
- Meromy, L., Molotch, N. P., Link, T. E., Fassnacht, S. R., & Rice, R. (2013). Subgrid variability of snow water equivalent at operational snow stations in the Western USA. *Hydrological Processes*, 27(17), 2383–2400. https://doi.org/10.1002/hyp.9355
- Miller, M. P., Buto, S. G., Susong, D. D., & Rumsey, C. A. (2016). The importance of base flow in sustaining surface water flow in the Upper Colorado River Basin. Water Resources Research, 52(5), 3547–3562. https://doi.org/10.1002/2015wr017963
- Miller, O. L., Putman, A. L., Alder, J., Miller, M., Jones, D. K., & Wise, D. R. (2021). Changing climate drives future streamflow declines and challenges in meeting water demand across the southwestern United States. *Journal of Hydrology X*, 11, 100074. https://doi.org/10.1016/j. hydroa.2021.100074
- Milly, P. C., & Dunne, K. A. (2020). Colorado River flow dwindles as warming-driven loss of reflective snow energizes evaporation. *Science*, 367(6483), 1252–1255. https://doi.org/10.1126/science.aay9187
- Moore, C., Kampf, S., Stone, B., & Richer, E. (2015). A GIS-based method for defining snow zones: Application to the Western United States. Geocarto International, 30(1), 62–81. https://doi.org/10.1080/10106049.2014.885089
- Moriasi, D. N., Arnold, J. G., Liew, M. W. V., Bingner, R. L., Harmel, R. D., & Veith, T. L. (2007). Model evaluation guidelines for systematic quantification of accuracy in watershed simulations. *Transactions of the ASABE*, 50(3), 885–900. https://doi.org/10.13031/2013.23153
- Mortezapour, M., Menounos, B., Jackson, P. L., & Erler, A. R. (2022). Future snow changes over the Columbia mountains, Canada, using a distributed snow model. *Climatic Change*, 172(1), 1–24. https://doi.org/10.1007/s10584-022-03360-9
- Mote, P. W., Li, S., Lettenmaier, D. P., Xiao, M., & Engel, R. (2018). Dramatic declines in snowpack in the Western US. npj Climate and Atmospheric Science, 1(1), 1–6. https://doi.org/10.1038/s41612-018-0012-1
- Musselman, K. N., Addor, N., Vano, J. A., & Molotch, N. P. (2021). Winter melt trends portend widespread declines in snow water resources. *Nature Climate Change*, 11(5), 418–424. https://doi.org/10.1038/s41558-021-01014-9
- Musselman, K. N., Clark, M. P., Liu, C., Ikeda, K., & Rasmussen, R. (2017). Slower snowmelt in a warmer world. Nature Climate Change, 7(3), 214–219. https://doi.org/10.1038/nclimate3225
- Musselman, K. N., Molotch, N. P., & Margulis, S. A. (2017). Snowmelt response to simulated warming across a large elevation gradient, southern Sierra Nevada, California. The Cryosphere, 11(6), 2847–2866. https://doi.org/10.5194/tc-11-2847-2017
- National Research Council. (2007). Colorado River Basin water management: Evaluating and adjusting to hydroclimatic variability. National Academies Press.
- Nowak, K., Hoerling, M., Rajagopalan, B., & Zagona, E. (2012). Colorado River Basin hydroclimatic variability. Journal of Climate, 25(12), 4389–4403. https://doi.org/10.1175/jcli-d-11-00406.1
- Perry, D. M., & Praskievicz, S. J. (2017). A new era of big infrastructure?(Re) developing water storage in the US west in the context of climate change and environmental regulation. *Water Alternatives*, 10(2), 437–454.
- Prasad, R., Tarboton, D. G., Liston, G. E., Luce, C. H., & Seyfried, M. S. (2001). Testing a blowing snow model against distributed snow measurements at Upper Sheep Creek, Idaho, United States of America. *Water Resource Research*, 37(5), 1341–1356. https://doi.org/10. 1029/2000WR900317
- Rasmussen, R., Baker, B., Kochendorfer, J., Meyers, T., Landolt, S., Fischer, A. P., et al. (2011). How well are we measuring snow: The NOAA/ FAA/NCAR winter precipitation test bed. *Bulletin America Meteorology Social*, 93(6), 811–829. https://doi.org/10.1175/BAMS-D-11-00052.1
- Rasmussen, R., & Liu, C. (2017). High resolution WRF simulations of the current and future climate of North America. Research Data Archive at the National Center for Atmospheric Research, Computational and Information Systems Laboratory. https://doi.org/10.5065/D6V40SXP
- Rasmussen, R., Liu, C., Ikeda, K., Gochis, D., Yates, D., Chen, F., et al. (2011). High-resolution coupled climate runoff simulations of seasonal snowfall over Colorado: A process study of current and warmer climate. *Journal of Climate*, 24(12), 3015–3048. https://doi.org/ 10.1175/2010jcli3985.1
- Reclamation (2016). SECURE water act section 9503(c)—reclamation climate change and water. U.S. Department of the Interior, Bureau of Reclamation.
- Reclamation (Bureau of Reclamation). (2021). Water reliability in the West—2021 SECURE water act report. Prepared for the United States congress. Bureau of Reclamation, Water Resources and Planning Office.
- Rhoades, A. M., Jones, A. D., & Ullrich, P. A. (2018). Assessing mountains as natural reservoirs with a multimetric framework. *Earth's Future*, 6(9), 1221–1241. https://doi.org/10.1002/2017ef000789
- Rhoades, A. M., Ullrich, P. A., & Zarzycki, C. M. (2018). Projecting 21st century snowpack trends in Western USA mountains using variableresolution CESM. *Climate Dynamics*, 50(1), 261–288. https://doi.org/10.1007/s00382-017-3606-0

Robles, M. D., Hammond, J. C., Kampf, S. K., Biederman, J. A., & Demaria, E. (2021). Winter inputs buffer streamflow sensitivity to snowpack losses in the salt river watershed in the lower Colorado River Basin. *Water*, 13(1), 3. https://doi.org/10.3390/w13010003

- Salzmann, N., & Mearns, L. O. (2012). Assessing the performance of multiple regional climate model simulations for seasonal mountain snow in the Upper Colorado River Basin. *Journal of Hydrometeorology*, 13(2), 539–556. https://doi.org/10.1175/2011jhm1371.1
- Sato, T., Kimura, F., & Kitoh, A. (2007). Projection of global warming onto regional precipitation over Mongolia using a regional climate model. Journal of Hydrology, 333(1), 144–154. https://doi.org/10.1016/j.jhydrol.2006.07.023
- Scalzitti, J., Strong, C., & Kochanski, A. (2016a). Climate change impact on the roles of temperature and precipitation in Western U.S. snowpack variability. *Geophysical Research Letters*, 43(10), 5361–5369. https://doi.org/10.1002/2016GL068798
- Scalzitti, J., Strong, C., & Kochanski, A. K. (2016b). A 26 year high-resolution dynamical downscaling over the Wasatch Mountains: Synoptic effects on winter precipitation performance. *Journal of Geophysical Research: Atmospheres*, 121(7), 3224–3240. https://doi.org/ 10.1002/2015jd024497
- Scanlon, B. R., Reedy, R. C., Faunt, C. C., Pool, D., & Uhlman, K. (2016). Enhancing drought resilience with conjunctive use and managed aquifer recharge in California and Arizona. *Environmental Research Letters*, 11(3), 035013. https://doi.org/10.1088/1748-9326/11/3/035013
- Schär, C., Frei, C., Lüthi, D., & Davies, H. C. (1996). Surrogate climate-change scenarios for regional climate models. *Geophysical Research Letters*, 23(6), 669–672. https://doi.org/10.1029/96gl00265

- Serreze, M. C., Clark, M. P., Armstrong, R. L., McGinnis, D. A., & Pulwarty, R. S. (1999). Characteristics of the Western United States snowpack from snowpack telemetry (SNOTEL) data. *Water Resource Research*, *35*(7), 2145–2160. https://doi.org/10.1029/1999WR900090
- Sexstone, G., Barnhart, T., & Hammond, J. (2022). *High resolution current and future climate SnowModel simulations in the upper Colorado River Basin*. USGS Sciencebase data release. https://doi.org/10.5066/P9K0QUCN
- Sexstone, G. A., Clow, D. W., Fassnacht, S. R., Liston, G. E., Hiemstra, C. A., Knowles, J. F., & Penn, C. A. (2018). Snow sublimation in mountain environments and its sensitivity to forest disturbance and climate warming. *Water Resource Research*, 54(2), 1191–1211. https://doi. org/10.1002/2017WR021172
- Sexstone, G. A., Driscoll, J. M., Hay, L. E., Hammond, J. C., & Barnhart, T. B. (2020). Runoff sensitivity to snow depletion curve representation within a continental scale hydrologic model. *Hydrological Processes*, 34(11), 2365–2380. https://doi.org/10.1002/hyp.13735
- Sexstone, G. A., & Fassnacht, S. R. (2014). What drives basin scale spatial variability of snowpack properties in northern Colorado? The Cryosphere, 8(2), 329–344. https://doi.org/10.5194/tc-8-329-2014
- Sexstone, G. A., Penn, C. A., Liston, G. E., Gleason, K. E., Moeser, C. D., & Clow, D. W. (2020). Spatial variability in seasonal snowpack trends across the Rio Grande headwaters (1984–2017). Journal of Hydrometeorology, 21(11), 2713–2733. https://doi.org/10.1175/jhm-d-20-0077.1
- Siirila-Woodburn, E. R., Rhoades, A. M., Hatchett, B. J., Huning, L. S., Szinai, J., Tague, C., et al. (2021). A low-to-no snow future and its impacts on water resources in the Western United States. *Nature Reviews Earth and Environment*, 2(11), 1–20. https://doi.org/10.1038/ s43017-021-00219-y
- Sproles, E. A., Nolin, A. W., Rittger, K., & Painter, T. H. (2013). Climate change impacts on maritime mountain snowpack in the Oregon Cascades. *Hydrology and Earth System Sciences*, 17(7), 2581–2597. https://doi.org/10.5194/hess-17-2581-2013
- Sterle, K., Jose, L., Coors, S., Singletary, L., Pohll, G., & Rajagopal, S. (2020). Collaboratively modeling reservoir reoperation to adapt to earlier snowmelt runoff. *Journal of Water Resources Planning and Management*, 146(1), 05019021. https://doi.org/10.1061/(asce)wr.1943-5452.0001136
- Stewart, I. T., Cayan, D. R., & Dettinger, M. D. (2005). Changes toward earlier streamflow timing across Western North America. Journal of Climate, 18(8), 1136–1155. https://doi.org/10.1175/jcli3321.1
- Sumargo, E., Cannon, F., Ralph, F. M., & Henn, B. (2020). Freezing level forecast error can consume reservoir flood control storage: Potentials for Lake Oroville and New Bullards Bar reservoirs in California. Water Resources Research, 56(8), e2020WR027072. https://doi.org/ 10.1029/2020wr027072
- Tague, C., & Peng, H. (2013). The sensitivity of forest water use to the timing of precipitation and snowmelt recharge in the California Sierra: Implications for a warming climate. *Journal of Geophysical Research: Biogeosciences*, *118*(2), 875–887. https://doi.org/10.1002/jgrg.20073
- Trujillo, E., & Molotch, N. P. (2014). Snowpack regimes of the Western United States. Water Resources Research, 50(7), 5611–5623. https:// doi.org/10.1002/2013wr014753
- Udall, B., & Overpeck, J. (2017). The twenty-first century Colorado River hot drought and implications for the future. *Water Resources Research*, 53(3), 2404–2418. https://doi.org/10.1002/2016wr019638
- U.S. Geological Survey. (2020). 3D elevation Program 1 arc-second resolution digital elevation model. Retrieved from https://www.usgs.gov/ the-national-map-data-delivery
- Varhola, A., Coops, N. C., Weiler, M., & Moore, R. D. (2010). Forest canopy effects on snow accumulation and ablation: An integrative review of empirical results. *Journal of Hydrology*, 392(3–4), 219–233. https://doi.org/10.1016/j.jhydrol.2010.08.009
- Westerling, A. L. (2016). Increasing Western US forest wildfire activity: Sensitivity to changes in the timing of spring. *Philosophical Transac*tions of the Royal Society B: Biological Sciences, 371(1696), 20150178. https://doi.org/10.1098/rstb.2015.0178
- Westerling, A. L., Hidalgo, H. G., Cayan, D. R., & Swetnam, T. W. (2006). Warming and earlier spring increase Western US forest wildfire activity. *Science*, 313(5789), 940–943. https://doi.org/10.1126/science.1128834
- Wi, S., Dominguez, F., Durcik, M., Valdes, J., Diaz, H. F., & Castro, C. L. (2012). Climate change projection of snowfall in the Colorado River Basin using dynamical downscaling. *Water Resources Research*, 48(5), W05504. https://doi.org/10.1029/2011wr010674
- Woodhouse, C. A., Gray, S. T., & Meko, D. M. (2006). Updated streamflow reconstructions for the Upper Colorado River Basin. Water Resource Research, 42(5), W05415. https://doi.org/10.1029/2005WR004455
- Wrzesien, M. L., & Pavelsky, T. M. (2020). Projected changes to extreme runoff and precipitation events from a downscaled simulation over the Western United States. Frontiers of Earth Science, 7, 355. https://doi.org/10.3389/feart.2019.00355
- Xu, D., Ivanov, V. Y., Li, X., & Troy, T. J. (2021). Peak runoff timing is linked to global warming trajectories. *Earth's Future*, 9(8), e2021EF002083. https://doi.org/10.1029/2021ef002083
- Yang, D., Goodison, B. E., Metcalfe, J. R., Golubev, V. S., Bates, R., Pangburn, T., & Hanson, C. L. (1998). Accuracy of NWS 8" Standard Nonrecording Precipitation Gauge: Results and application of WMO Intercomparison. *Journal of Atmospheric and Oceanic Technology*, 15(1), 54–68. https://doi.org/10.1175/1520-0426(1998)015<0054:AONSNP>2.0.CO;2
- Zhang, X. (2015). Conjunctive surface water and groundwater management under climate change. *Frontiers in Environmental Science*, *3*, 59. https://doi.org/10.3389/fenvs.2015.00059